Author's response to reviews

Title: Local problems; local solutions: An innovative approach to investigating and addressing causes of maternal deaths in Zambia's Copperbelt.

Authors:

mary b hadley (marybhadley@hotmail.com)
mary tuba (mary_tuba2004@yahoo.co.uk)

Version: 6 Date: 12 May 2011

Author's response to reviews:

boxes changed to tables and put at end of manuscript
vertical lines removed from tables
figures comply with guidelines